Author's response to reviews

Title: Silent Cerebral Infarct After Cardiac Catheterization as Detected by Diffusion Weighted Magnetic Resonance Imaging: A Randomized Comparison of Radial and Femoral arterial approaches.

Authors:

Michele Hamon (hamon-mi@chu-caen.fr)
Francesco Burzotta (f.burzotta@eudoramail.com)
Catherine Oppenheim (oppenheim-C@APHP.stanne.fr)
Remy Morello (morello-r@chu-caen.fr)
Fausto Viader (viader-f@chu-caen.fr)
Martial Hamon (hamon-m@chu-caen.fr)

Version: 3 Date: 12 December 2006

Author's response to reviews:

Dear Editor,

According to recommendations we have reformatted the manuscript with a design and methods section including two additional figures for clarification.

We have also added a discussion section as requested and an appendix with all committees of the study and investigators up today.

All authors of the manuscript have approved final version and contribute to the design and substantially in writing and editing the manuscript.

looking forward to hear from you

Best regards

Martial Hamon, MD